

<mark>-</mark>

Water Resources Research

RESEARCH ARTICLE

10.1029/2020WR027296

Key Points:

- Simulated annual maximum and summer median lake chlorophyll-*a* are positively associated with external phosphorus load and air temperature, respectively
- A 2°C rise in simulated air temperature significantly increased summer median, but not annual maximum, chlorophyll-a
- Annual maximum chlorophyll-a may be more informative than median summer chlorophyll-a to assess nutrient load effects on water quality

Supporting Information:

Supporting information S1

Correspondence to: N. K. Ward,

nkward@vt.edu

Citation:

Ward, N. K., Steele, B. G., Weathers, K. C., Cottingham, K. L., Ewing, H. A., Hanson, P. C., & Carey, C. C. (2020). Differential responses of maximum versus median chlorophyll-*a* to air temperature and nutrient loads in an oligotrophic lake over 31 years. *Water Resources Research*, 56, e2020WR027296. https://doi.org/ 10.1029/2020WR027296

Received 8 FEB 2020 Accepted 22 MAY 2020 Accepted article online 28 MAY 2020

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

Differential Responses of Maximum Versus Median Chlorophyll-*a* to Air Temperature and Nutrient Loads in an Oligotrophic Lake Over 31 Years

Nicole K. Ward¹, Bethel G. Steele², Kathleen C. Weathers², Kathryn L. Cottingham³, Holly A. Ewing⁴, Paul C. Hanson⁵, and Cayelan C. Carey¹

¹Virginia Tech, Department of Biological Sciences, Blacksburg, VA, USA, ²Cary Institute of Ecosystem Studies, Millbrook, NY, USA, ³Department of Biological Sciences, Dartmouth College, Hanover, NH, USA, ⁴Program in Environmental Studies, Bates College, Lewiston, ME, USA, ⁵Center for Limnology, University of Wisconsin-Madison, Madison, WI, USA

Abstract Globally, phytoplankton abundance is increasing in lakes as a result of climate change and land-use change. The relative importance of climate and land-use drivers has been examined primarily for mesotrophic and eutrophic lakes. However, oligotrophic lakes show different sensitivity to climate and land-use drivers than mesotrophic and eutrophic lakes, necessitating further exploration of the relative contribution of the two drivers of change to increased phytoplankton abundance. Here, we investigated how air temperature (a driver related to climate change) and nutrient load (a driver related to land-use and climate change) interact to alter water quality in oligotrophic Lake Sunapee, New Hampshire, USA. We used long-term data and the one-dimensional hydrodynamic General Lake Model (GLM) coupled with Aquatic EcoDyanmics (AED) modules to simulate water quality. Over the 31-year simulation, summer median chlorophyll-a concentration was positively associated with summer air temperature, whereas annual maximum chlorophyll-a concentration was positively associated with the previous 3 years of external phosphorus load. Scenario testing demonstrated a 2°C increase in air temperature significantly increased summer median chlorophyll-a concentration, but not annual maximum chlorophyll-a concentration. For both maximum and median chlorophyll-a concentration, doubling external nutrient loads of total nitrogen and total phosphorus at the same time, or doubling phosphorus alone, resulted in a significant increase. This study highlights the importance of aligning lake measurements with the ecosystem metrics of interest, as maximum chlorophyll-a concentration may be more uniquely sensitive to nutrient load and that typical summer chlorophyll-a concentration may increase due to warming alone.

Plain Language Summary Clear water lakes are experiencing more frequent water quality problems due to land development and climate change. However, it is challenging to identify how land development and climate change interact to alter water quality because their effects are complex and occurring at the same time. We used three decades of observational data combined with a lake ecosystem simulation model to explore the role of land development and climate change on water quality. Our water quality indicator of focus was phytoplankton, which are small photosynthesizing organisms in the water, often referred to as "algae." We found that the effects of land use and climate depend on if we look at yearly maximum or average phytoplankton concentrations. Average phytoplankton concentrations during the summer (representing typical summer conditions) increase with either warmer air temperatures or higher nutrient pollution. However, annual maximum phytoplankton concentration (representing phytoplankton swith higher nutrient pollution. Typical summer phytoplankton concentrations will likely increase with warmer air temperatures due to climate change alone and increase even further when combined with higher nutrient pollution. To maintain clear water lakes, nutrient pollution should be reduced even more than previously thought to compensate for increasing phytoplankton in a warmer climate.

1. Introduction

Lake water quality is often considered to be most influenced by catchment land-use change, whereas climate-induced changes are of secondary importance (Brookes & Carey, 2011; Bucak et al., 2018; Couture et al., 2014, 2018; Jenny et al., 2015). However, since lake water quality is altered through multiple interact-ing land-use and climate change pathways (Figure 1), it is difficult to tease apart the relative contribution of





Figure 1. Pathways for climate and land use to affect phytoplankton: (1) the Air Temperature Pathway may either stimulate or inhibit phytoplankton growth, for example, warmer surface water may stimulate phytoplankton growth, but earlier onset of thermal stratification may lower nutrient exchange with the hypolimnion, inhibiting phytoplankton growth; (2) the Precipitation Pathway increases stream discharge (potentially increasing overall nutrient load) and increases transport of nutrients from terrestrial landscape (increasing in-stream nutrient concentrations and overall nutrient load); and (3) the Catchment Nutrient Pathway changes nutrient availability and transport from terrestrial landscape (e.g., fertilizer application increases nutrient concentrations in inflow streams; land use development increases impervious surface and erosion). "[Nutrient]" denotes nutrient (nitrogen or phosphorus) concentrations; solid arrows indicate shorter time-scale relationships, and dashed arrows indicate lagged effects as nutrients accumulate in the catchment before being transported to the lake.

each individual driver to overall water quality (Collins et al., 2019; McCullough et al., 2019; Moss et al., 2011). Land-use and land-cover change (e.g., fertilizer application and deforestation) can increase nutrient loads from catchments to lakes through increased stream nutrient concentrations (Bormann et al., 1974; King et al., 2007) via the Catchment Nutrient Pathway (Figure 1). Changes in precipitation frequency or intensity due to climate change can mobilize more terrestrial nutrients and increase stream discharge, consequently increasing total nutrient loads to the lake (Arvola et al., 2015; Jeppesen et al., 2009; Meyer et al., 1999) via the Precipitation Pathway (Figure 1). Additionally, rising air temperatures due to climate change increase surface water temperature, thereby affecting lake biological activity, the duration of summer stratification, and the frequency and duration of hypoxia (De Senerpont Domis et al., 2013; Magee et al., 2019; O'Reilly et al., 2015; Paerl & Huisman, 2008; Woolway et al., 2019) via the Air Temperature Pathway (Figure 1). Due to the interaction between drivers, it is thus possible that altering one pathway may amplify or dampen the lake response and sensitivity to another pathway.

Phytoplankton biomass, and especially its key proxy, chlorophyll-*a* concentration, is useful in assessing lake ecosystem responses to the combined effects of land-use and climate change, since increased nutrients and water temperatures can directly stimulate phytoplankton (Carey, Ibelings, et al., 2012; Trolle et al., 2014; Williamson et al., 2008). As chlorophyll-*a* concentrations increase, there is more variability in the ratio of summer mean versus maximum chlorophyll-*a* concentration (Jones et al., 1979). The increased variability of the ratio of mean-to-maximum chlorophyll-*a* concentration could be because central tendency and maximum chlorophyll-*a* concentration represent different ecosystem responses. Central tendency (e.g., mean or median) chlorophyll-*a* concentration may correspond to typical conditions in the lake, similar to what

would be used to determine trophic state (Carlson, 1977). In contrast, annual maximum chlorophyll-*a* concentration may reflect short-term increases in phytoplankton as a result of a large nutrient loading event or a period of favorable temperature conditions (e.g., Batt et al., 2017). Thus, different summary statistics (e.g., median vs. maximum) of the distribution of chlorophyll-*a* may reveal unique signals of change and may help elucidate the role of different pathways of change in lake ecosystems (Figure 1).

To date, most work assessing the effect of climate and land-use change on phytoplankton biomass or chlorophyll-*a* concentrations has focused on eutrophic lakes and reservoirs, but some recent studies including oligotrophic lakes indicate differential sensitivity to climate and land-use drivers (Hayes et al., 2015; Rigosi et al., 2014). For example, high-nutrient lakes situated in predominantly agricultural catchments can shift from P-limited to N-limited under drought conditions, in contrast to lakes in forested catchments, which tend to be N-limited during drought or nondrought conditions (Hayes et al., 2015). Further, oligotrophic lakes in the northeastern United States are more sensitive to air temperature and precipitation changes than lakes in the same region with higher productivity (McCullough et al., 2019). Given increasing reports of cyanobacterial blooms in oligotrophic lakes in northeastern North America (e.g., Carey et al., 2012; Pick, 2015; Winter et al., 2011) and the rapid loss of oligotrophic freshwater systems (Stoddard et al., 2016), teasing apart the effects of climate and land-use change on oligotrophic lakes is particularly timely.

Disentangling the relative role of climate and land use is challenging, given the interaction of different pathways (Figure 1); however, simulation modeling, when used in conjunction with observational data collected across a range of environmental conditions, may further our understanding of the long-term role of multiple drivers on ecosystem processes (Turner & Carpenter, 2017). For example, in oligotrophic Lake Sunapee (NH, USA), chlorophyll-*a* concentrations are generally low, with occasional brief increases (Carey, Ewing, et al., 2012; Carey, Weathers, et al., 2014). Multiple limitations, such as difficulty measuring low concentrations of chlorophyll-*a* with precision, intermittent field monitoring programs that may miss extreme events, and a complicated suite of physical-biological processes that influence the observed temporal patterns of chlorophyll-*a* concentration, contribute to difficulty in assessing multiple drivers in ecosystem analysis (Weathers et al., 2013). However, a well-calibrated simulation model enables us to explore the role of each driver, because environmental drivers can be manipulated independently and their effects on chlorophyll-*a* concentration can be examined through mechanisms instantiated in the model. Simulation models can also provide data at very high temporal resolution relative to empirical data, enabling comparisons of different types of summary statistics—such as central tendencies versus extremes—that are rarely possible with manual field measurements alone.

We evaluated the role of terrestrial nutrient loading and climate on simulated annual maximum and summer median chlorophyll-*a* concentrations by asking two questions: (1) How do simulated annual maximum and summer median chlorophyll-*a* concentrations respond to climate and nutrient-load drivers in an oligotrophic lake? (2) How do simulated annual maximum and summer median chlorophyll-*a* concentrations in an oligotrophic lake respond to projected increases in air temperature and stream inflow nutrient concentration? We used historical climate, stream, and lake monitoring data to calibrate a one-dimensional hydrodynamic model coupled with an aquatic ecosystem model to produce a 31-year reference simulation of water quality in oligotrophic Lake Sunapee, NH to address Question 1. We then applied multiple factorial scenarios of elevated stream nutrient concentrations and air temperature to the calibrated model to address Question 2.

2. Materials and Methods

2.1. Study Site Description

Lake Sunapee is located on the border of Merrimack and Sullivan counties in New Hampshire, USA ($43^{\circ}24'$ N, $72^{\circ}3'W$). Lake Sunapee is the sixth largest lake in New Hampshire and is an important drinking water source and recreation area (Sunapee Area Watershed Coalition, 2008). Lake Sunapee has a surface area of 16.55 km², a volume of 1.88×10^8 m³, residence time of 3.1 years, mean depth of 10 m, maximum depth of 33 m, and maximum fetch of 9.1 km (Carey, Weathers, et al., 2014; Richardson et al., 2017). The lake is dimictic, with ice cover from December or January until late March or April (Bruesewitz et al., 2015), and a mean summertime thermocline maximum depthof 6–8 m (Carey, Weathers, et al., 2014).





Lake Sunapee, New Hampshire, U.S.A.

Figure 2. Location and bathymetry of Lake Sunapee, New Hampshire, USA. Bathymetric map source: Lake Sunapee Protective Association.

The Lake Sunapee Protective Association (LSPA; lake-sunapee.org) is a member of the Global Lake Ecological Observatory Network (GLEON; gleon.org), and deploys and maintains a monitoring buoy located near the deepest point in the lake (Figure 2). The buoy has sensors that measure meteorological data, a water temperature depth profile, and epilimnetic dissolved oxygen concentration at 10- to 15-minute intervals (all data available in Environmental Data Initiative Repository; LSPA et al., 2018; Richardson et al., 2020).

Lake Sunapee is an oligotrophic, clear-water lake with historical chlorophyll-*a* concentrations less than 5 μ g L⁻¹ (Steiner & Titus, 2017; NHDES, 2017) and dissolved organic carbon concentrations less than 2.5 mg L⁻¹ (Solomon et al., 2013). The surrounding catchment is approximately 80% forested, 7% wetlands and ponds, 6% developed, 4% agricultural land, and 3% miscellaneous open space, but has been facing increasing development pressure (Cobourn et al., 2018; CSRC 2002; Yang et al., 2018), especially around the perimeter of the lake. The lake has six major inflows, which are estimated to collectively account for roughly 90% of all surface water inputs to the lake (Schloss, 1990). Lake Sunapee air temperature has been increasing at a rate of 0.42°C per decade since 1979 (*p* value <0.001; Figure S1 in the supporting information).



2.2. Model Description and Driver Data

We used the General Lake Model (GLM) v.2.1.8, a one-dimensional hydrodynamic model coupled with Aquatic EcoDynamics (AED), a lake ecosystem model (Hipsey et al., 2019). GLM and AED are open-source and are available online (http://aed.see.uwa.edu.au/research/models/GLM/%20 and https:// github.com/AquaticEcoDynamics/GLM).

GLM-AED requires many different driver data files and configuration files (Hipsey et al., 2019), including initial water chemistry and temperature, daily stream inflow temperature, volume, and nutrient concentrations (see below), daily lake outflow volume (Ward et al., 2020), and hourly weather. Hourly weather data at the buoy site (see section 2.1 and Figure 2) from 1982 to 2015 were downloaded from the North American Land Data Assimilation System (NLDAS-2) website (Xia et al., 2012). We estimated daily stream inflow volume to the lake based on precipitation from NLDAS-2 and a simple overland flow relationship based on a runoff coefficient and impervious surface area (held constant through the simulation), as well as a snow accumulation and snowmelt model (McCarthy, 2008; USCOE, 1998; Van Mullem et al., 2004). Base flow was added to the precipitation-generated inflow such that their sum produced a total flux sufficient to reproduce the known mean residence time of the lake (3.1 years) (Richardson et al., 2017; Schloss, 1990), given the lake volume. Daily inflow water temperature was modeled using two linear regressions with NLDAS-2 air temperature and stream temperature data (Ewing et al., 2020), one for cold (subzero) air temperatures and one for air temperatures above 0°C (Text S1 and Figures S2–S9). Observed stream temperature and final stream inflow volume and temperature driver data are available in the Environmental Data Initiative repository (Ewing et al., 2020; Ward et al., 2020).

GLM-AED driver data also include daily nutrient concentrations for inflow streams, representing terrestrial and stream nutrient contributions to the lake. Inflow chemistry data were obtained from LSPA's long-term monitoring program, in which measurements of total phosphorus (TP) were available for inflow streams at monthly intervals each summer from 1986–2015 (*n* of observed values = 2,692 from 1986 to 2015). Although stream inflow concentrations often co-vary with stream inflow volume, there was no volume-concentration relationship in our observations, so our modeled daily stream concentration was independent of stream inflow volume. On days without observed inflow stream TP concentration, we used the *rnorm* function in R (R version 3.3.3, R Core Team, 2017) to randomly sample with replacement from a randomly generated normal distribution (n = 1,000, $\mu = 5$ -year moving window mean of observed values, $\sigma = 5$ -year moving window variance of observed values); we used the 5-year distribution of observations from 1986 to 1991 to bootstrap the years prior to 1986. TP concentrations were divided into fractions based on observations from oligotrophic Ontario Lakes with similar trophic status, bedrock, and land use (Wetzel, 2001): 50% adsorbed P, 2.95% soluble reactive P, 32.7% particulate organic P, and 14.35% dissolved organic P.

On days without observed total nitrogen (TN) inflow stream concentrations, we created a TN:TP mass ratio for every day using the *rnorm* function in R to randomly sample with replacement from a randomly generated normal distribution of TN:TP (n = 1,000, $\mu =$ mean of observed TN:TP values, $\sigma =$ variance of observed TN:TP values; n of observed values = 135, from 2009 to 2012) and then back-calculated TN using the randomly generated TN:TP and previously estimated TP value. In lieu of observational data, we divided TN concentrations into fractions of particulate organic N (40%), dissolved organic N (40%), nitrate (10%), and ammonium (10%), following literature values (reviewed by Wetzel, 2001) and assuming that available N would be lower than organic N pools. Daily dissolved organic carbon (DOC) and particulate organic carbon (POC) values were held constant at 1.4 g L⁻¹ and 0.15 g L⁻¹, respectively, throughout the 31-year simulation (n = 132 from 2004, 2005, and 2009–2013) due to the low number of observations and limited direct comparisons for generating a total organic carbon:TP relationship. Stream nutrient concentration driver data are available in the Environmental Data Initiative repository (Ward et al., 2020).

2.3. Model Calibration and Evaluation

The total simulation period for the Lake Sunapee GLM-AED model was 15 April 1982 to 31 December 2015. The simulation has four distinct periods: (1) a spin-up period from 15 April 1982 to 31 December 1984, (2) a calibration period from 1 January 2005 to 31 December 2009, (3) an evaluation period from 1 January 2010 to 31 December 2013, and (4) the periods of both 1 January 1985 to December 2004 and 1 January 2014 to 31 December 2015 in which no goodness-of-fit metrics were calculated due to limited observational data. We

chose these time periods for calibration and evaluation because they cover a wide range in annual temperature and precipitation and coincide with a time period of sufficient available data for comparison. We ran the model on an hourly time step, simulating vertical whole-water profile conditions at the deepest location of the lake.

Before initializing the full 31-year simulation, we conducted a shorter simulation for calibration of model parameters and evaluation of model fit. We focused on 2005 through 2009 for the calibration phase, which included a range of warm, cool, dry, and wet years. To account for potential effects of initial conditions, we began analyzing model outputs after the simulation ran through one full summer stratified and fall turnover period. Specifically, for the model calibration process, we initiated the model simulation in April 2004, setting the initial water chemistry profile to observed values measured at the buoy site of the lake in April 2004, but we did not use model outputs for calibration until 1 January 2005. Examining model outputs from 2005 through 2009, we systematically adjusted lake physics, nutrient cycling, and phytoplankton growth parameters (Table S1) to improve the model fit with available observed data based on goodness-of-fit metrics (following Bennett et al., 2013; for final parameters and initial condition data sets, see Ward et al., 2020). Due to the coarse temporal resolution of observed P concentrations and chlorophyll-a concentration data, we calibrated the model to represent expected annual patterns in P and chlorophyll-a concentration in addition to comparing model outputs with infrequent observations. For example, we ensured cyclical patterns of chlorophyll-a concentration were recreated in the model (e.g., peaks in late summer, decreases during turnover and ice-cover periods). Then, we fine-tuned the magnitude of chlorophyll-a during the summer stratified period using direct comparison with observational data (most of which were collected during the summer stratified period and did not show discernible patterns within each year). For nitrogen, we calibrated the model to anticipated annual cyclical patterns and approximated the magnitude of TN to observed pelagic TN (Cottingham, 2020) and the relative proportion of each constituent to be in agreement with expected ratios in oligotrophic lakes (Wetzel, 2001). During the calibration period (2005-2009), we were only able to make direct comparisons of simulated and observed pelagic N in 2009, due to there being no observational measurements of pelagic N earlier in the calibration period. The simulated TN concentrations were within the range of observed pelagic TN concentrations (Carey, Cottingham, et al., 2014; Cottingham, 2020). Conversely, we calibrated the model to daily values for temperature and dissolved oxygen, as we had high-frequency observations of these variables beginning in 2007 (Richardson et al., 2020).

After final parameter values were established, we calculated the same goodness-of-fit metrics during the 2010-2013 evaluation period. This period was selected for evaluation due to there being adequate observational data for comparisons. Goodness-of-fit metrics were calculated for model output and high-frequency observations of temperature and dissolved oxygen at noon each day during the calibration and evaluation period. Due to limited frequency of observational data for TP, TN, and chlorophyll-a concentration, we calculated the same goodness-of-fit metrics used on high-frequency comparisons and compared observed and simulated medians and ranges for the days of observation with expected concentrations in oligotrophic lakes. All goodness-of-fit metrics were calculated using the hydroGOF package (Zambrana-Bigiarini, 2017) in R. These included R^2 (best fit = 1), root mean square error (RMSE; which presents model error in the same units as the variable of interest, best fit = 0, percent bias (PBIAS; a measure of bias in predictions with overpredictions positive and underpredictions negative as a % of the magnitude of observations, best fit = 0), Nash-Sutcliffe efficiency (NSE; a high-frequency comparison of model output and observations commonly used in discharge modeling, in which the best fit = 1, a value of 0 indicates the model predictions are as accurate as the mean of observed values, and a negative value indicates the mean of observed values is more accurate than the modeled values), and mean absolute error (MAE; in units of the variable of interest, best fit = 0).

2.4. Reference Simulation Analysis—Question 1: How Do Annual Maximum Versus Median Chlorophyll-*a* Concentrations Respond to Climate and Land-Use Drivers in an Oligotrophic Lake?

To identify which drivers in the Catchment Nutrient Pathway, Precipitation Pathway, and Air Temperature Pathway (Figure 1) were most associated with interannual variability in chlorophyll-*a* concentrations, we assessed model outputs and driver data from the long-term Lake Sunapee GLM-AED simulation. Drivers within each of the three pathways are often interlinked, however, the goal here was to narrow the focus of driver variables, with further exploration of the pathways themselves in section 2.6. We used the final

parameters established during calibration (Ward et al., 2020), initialized the simulation on 15 April 1982, and ran the model through 31 December 2015. The purpose of the 31-year simulation was to explore how the model outputs change across a range of annual conditions. The relationship between drivers and chlorophyll-*a* concentration is likely complex, but for a first step at understanding the interactive effects of climate and land-use pathways on seasonal phytoplankton responses, we summarized daily model outputs to the seasonal scale. We refer to the initial simulation, aimed at reproducing lake conditions over the past 31 years, as the "reference" simulation rather than "baseline" in recognition that the lake is already undergoing ecosystem change (Richardson et al., 2017).

To assess whether different summary statistics of summer chlorophyll-*a* concentration were associated with different drivers as described above, we focused our analysis on annual maximum and summer median chlorophyll-*a* concentration. "Summer" was considered to be from 1 June 1 to 30 September each year. We selected the June through September window since this time period most consistently included summer stratification over the 31 years. Since we predicted that annual maximum and summer median chlorophyll-*a* concentration may respond to different drivers, we analyzed them separately, though they are correlated with each other (Spearman rho = 0.47, p < 0.01; Figure S10).

We used a multistep analysis to determine the association between land-use and climate drivers on summer median and annual maximum chlorophyll-*a* concentrations. We grouped potential driver variables into three categories: variables related to the (1) Precipitation Pathway (Figure 1; precipitation), (2) Air Temperature Pathway (air temperature), and (3) combined Catchment Nutrient Pathway and Precipitation Pathways (TN and TP loads) to identify the most associated driver variables within each pathway. For Question 1, we did not assess nutrient concentration separately from load because nutrient concentrations are made ecologically meaningful in combination with discharge, in the form of total nutrient load to the lake (e.g., Carpenter et al., 2018). We examined the role that increases in nutrient concentration in inflow streams could have on chlorophyll-*a* concentration response in Question 2.

We then aggregated each driver to spring (March through May), summer (June through September), water year (previous calendar year 1 October to current calendar year 30 September), and the previous three water years to assess more immediate versus lagged responses in chlorophyll-*a* concentration. The 3-year aggregation was selected due to the 3.1-year hydraulic residence time in the lake (Richardson et al., 2017; Schloss, 1990). We assessed different summary statistics for air temperature, including mean, minimum, and maximum, and used cumulative totals for TN and TP load and precipitation for each time period to identify possible driver-response associations.

We used Spearman correlations to identify which potential driver variables were the most strongly associated with summer median and maximum chlorophyll-*a* concentration. Annual summer median chlorophyll-*a* concentrations can be treated as independent from year to year, though there were weak lag-1 autocorrelations for maximum chlorophyll-*a* concentration that may affect inferences (Figures S11, S12). Nonparametric Spearman correlations were used to account for nonlinearity and potential outliers that bias Pearson correlation estimates (r).

2.5. Scenario Analysis—Question 2: How Do Maximum Versus Median Chlorophyll-*a* Concentrations in an Oligotrophic Lake Respond to Projected Increases in Air Temperature and Stream Inflow Nutrient Concentration?

To examine the relative contribution of higher stream inflow nutrient concentrations (Catchment Nutrient Pathway) and higher air temperatures (Air Temperature Pathway) to future chlorophyll-*a* concentrations, we generated five scenarios exploring different combinations of changes in air temperature and stream N and P concentrations that are plausible for the Lake Sunapee catchment (Table 1). The five scenarios were applied to the reference 1985–2015 simulation, in which the stream nutrient (TN and TP) concentrations (Catchment Nutrient Pathway, Figure 1; Scenarios 1, 2, 3, and 5, Table 1) and air temperature (Air Temperature Pathway, Figure 1; Scenarios 4 and 5, Table 1) driver data were manipulated, but all other conditions (e.g., stream discharge) were held constant.

The five hindcasted scenarios (Table 1) were based on expected future air temperature and inflow stream nutrient conditions for the Lake Sunapee catchment according to downscaled climate projections (Abatzoglou, 2013; Taylor et al., 2012) and economic development projections (Sunapee Area Watershed



Table 1

GLM-AED Hindcasted Scenario Descriptions

Name	Description	Driver data modification		
Scenario 1: TN \times 2	Increased inflow stream N concentration	Concentration of each fraction of N was doubled in comparison to reference simulation		
Scenario 2: TP \times 2	Increased inflow stream P concentration	Concentration of each fraction of P was doubled in comparison to reference simulation		
Scenario 3:	Increased inflow stream N and P concentrations	Concentration of each fraction of N and P was doubled in comparison to reference		
$TNTP \times 2$		simulation		
Scenario 4: +2°C	Increased air temperature	Air temperature was increased by 2°C in comparison to reference simulation		
Scenario 5:	Increased inflow stream N and P concentrations	Concentration of each fraction of N and P was doubled and air temperature was		
$TNTP \times 2 + 2^{\circ}C$	and air temperature	increased by 2°C in comparison to reference simulation		

Coalition, 2008). In Scenario 1: $TN \times 2$, Scenario 2: $TP \times 2$, and Scenario 3: $TNTP \times 2$, we doubled the daily concentration of each fraction of N and P, to result in two times the total N, P, or N and P load, respectively. In Scenario 4: $+2^{\circ}C$ scenario, we increased each hourly air temperature in the meteorological driver data file by 2°C. Finally, we increased air temperature and inflow stream N and P concentrations in Scenario 5: $TNTP \times 2 + 2^{\circ}C$, in which the concentration of each fraction of N and P was doubled, thereby doubling the total N and P load, and added 2°C to each hourly air temperature in the meteorological driver data. These scenarios were simplistic for the purpose of identifying the potential role of individual drivers in lake ecosystem outcomes. We recognize that changes to these drivers in reality will be more complex (e.g., nutrient loads will not increase uniformly, but likely will be characterized by more extreme peaks and higher background concentrations); however, for model simulations to elucidate lake process dynamics, simplified scenarios can be a useful tool (e.g., Piccolroaz et al., 2015).

The +2°C scenario represents additional temperature warming above the already-observed 1.4°C warming that occurred in the Sunapee catchment from 1981 to 2015 (Figure S1), representing a total of 3.4°C warming over reference 1981 air temperatures by the end of the simulation period. This 2°C warming above current conditions is expected at Lake Sunapee by 2050 according to the RCP 8.5 prediction, and by 2066 in the RCP 4.5 prediction, averaged across downscaled global climate models (Abatzoglou, 2013; Taylor et al., 2012). The TPTN \times 2 scenario is within the range of expected changes to inflow stream nutrient concentration in the catchment given local zoning and development (Sunapee Area Watershed Coalition, 2008).

We ran each scenario from 1982 through 2015 and assessed differences between the 31-year distribution of annual maximum and summer median chlorophyll-*a* concentration. We compared 31-year annual distributions of annual maximum and summer median chlorophyll-*a* concentrations to assess the effects of each scenario across a range of weather and loading conditions using Anderson-Darling tests (Razali & Wah, 2011) with a Bonferroni correction in R (R version 3.3.3, R Core Team, 2017).

2.6. Confirmation of Conceptual Driver Pathways With Scenario Model Output

To explicitly link the detailed model outputs with the conceptual driver pathways (Figure 1), we examined the relationships between the driver variables identified in the reference analysis (section 2.4), mediating variables (e.g., epilimnetic nutrient concentration and phytoplankton nutrient uptake), and the response variables (summer median and annual maximum chlorophyll-*a* concentration) across scenarios (section 2.5). The goal of this analysis was to leverage results from both Question 1 and Question 2 to determine if the driver variables associated with the Air Temperature Pathway and Catchment Nutrient Pathway (Figure 1) that were identified as important in the reference simulation analysis actually influenced annual and summer chlorophyll-*a* concentrations in the model as conceptualized in Figure 1. For this analysis, we examined relationships between driver data variables (e.g., nutrient loads), mediating variables that were simulated in the model (i.e., the variables that responded to changes in the driver variables and altered our focal response variables; e.g., epilimnetic P concentration and phytoplankton P uptake), and simulated focal response variables (annual maximum and summer median chlorophyll-*a* concentration) with Spearman correlations.

3. Results

3.1. Model Calibration and Evaluation

The GLM-AED model simulated water temperature and dissolved oxygen dynamics during the calibration (2005–2009) and evaluation (2010–2013) phase that closely tracked the observed data (Figures 3a and 3b).





Figure 3. Simulated and observed surface (0–6 m) mean water temperature (a); simulated and observed dissolved oxygen 1 m below the surface (b); simulated and observed surface (0–6 m) mean total phosphorus (c); simulated and observed surface (0–6 m) mean total nitrogen (d); and simulated and observed surface (0–3 m) mean chlorophyll-*a* (e). The white panel background is the calibration phase and gray panel background is the evaluation phase (for goodness-of-fit metrics, see Table 2). The method detection limit for total phosphorus observations in panel (c) was 5 μ g L⁻¹.

The goodness-of-fit metrics for high-frequency sensor data comparisons (water temperature and dissolved oxygen; Table 2, Figure 3) were within the range of previous GLM-AED simulations on north temperate lakes (e.g., Kara et al., 2012; Snortheim et al., 2017). In both simulated and observed data, surface water temperature warmed to ~25°C in the summer and cooled to 0°C in the winter (Figure 3a). Simulated dissolved oxygen at 1 m decreased in the summer to ~8.5 mg L⁻¹ and increased in the winter to ~12.5 mg L⁻¹ (Figure 3b).

GLM-AED simulated mean epilimnetic phosphorus, nitrogen, and chlorophyll-a concentrations during the calibration (2005-2009) and evaluation (2010-2013) phase were within the range expected for oligotrophic lakes (Figures 3c, 3d, and 3e). Given the low number of field observations for epilimnetic TP, TN, and chlorophyll-a concentrations, as well as the absence of subseasonal patterns in observational data, we compared medians and ranges of simulated and observed TP, TN, and chlorophyll-a concentrations on the days of observation in addition to the goodness-of-fit metrics (Table 2). Simulated epilimnetic TP (Figure 3c; 5.1–6.2 μ g L⁻¹; median = 5.6 μ g L⁻¹, n = 40) was similar to observed epilimnetic TP (5.0-7.0 μ g L⁻¹; median = 5.0 μ g L⁻¹, n = 40), and simulated epilimnetic TN (Figure 3d; 143–170 μ g L⁻¹; median = 162 μ g L⁻¹, n = 21) was similar to observed epilimnetic TN (127–191 μ g L⁻¹; median = 167 μ g L⁻¹, n = 21). Simulated surface chlorophyll-*a* concentration (Figure 3e; 0.1–2.2 μ g L⁻¹; median = 0.9 μ g L⁻¹, n = 41) was less than observed surface chlorophyll-*a* concentration (0.2–5.3 μ g L⁻¹; median = 1.64 μ g L⁻¹, n = 41). Overall, the model represented oligotrophic lake conditions well (e.g., mean chlorophyll- $a < 2.6 \ \mu g \ L^{-1}$ and TP <12 $\mu g \ L^{-1}$, Carlson, 1977; Figure 3). The model adequately represented the central tendencies of observed chlorophyll-a concentration but did not predict the peaks and slightly overpredicted TP (Table 2). The C:N in the lake was approximately 1:8 throughout the simulation, with particulate N composing the largest fraction of TN, similar to observations of littoral TN in Lake Sunapee. We note that our observed TP concentrations were very low (often below laboratory method detection limits and limits of quantitation).



Table 2

Goodness-of-Fit Metrics for Comparing Observed and Simulated GLM-AED Data for Lake Sunapee, NH, USA: Coefficient of Determination From Linear Regression (R^2) , Root Mean Square Error (RMSE), Percent Bias (PBIAS), Nash-Sutcliffe Efficiency (NSE), Mean Absolute Error (MAE); and n is the Number of Observed Measurements

Variable	Observation	Model period	п	R^2	RMSE	PBIAS	NSE	MAE
Temperature (°C) (0–6 m)	Buoy (daily max)	Calibration	781	0.99	1.25	6.0	0.98	1.05
		Evaluation	480	0.97	0.84	2.4	0.96	0.72
	Manual	Calibration	26	0.81	1.5	1.6	0.69	1.21
		Evaluation	14	0.85	1.34	1.3	0.74	1.10
Dissolved Oxygen (mg L^{-1}) (1 m)	Buoy (daily mean)	Calibration	415	0.57	0.43	-0.4	0.51	0.36
		Evaluation	278	0.57	0.42	0.4	0.48	0.34
	Manual	Calibration	26	0.07	1.34	1.9	0	0.77
		Evaluation	21	0.53	0.67	5.1	-0.04	0.51
Total P (μ g L ⁻¹) (0–6 m)	Manual	Calibration	26	0.02	0.64	8.2	-1.62	0.55
		Evaluation	14	0.14	0.69	6.0	-0.11	0.64
Total N (μ g L ⁻¹) (0–6 m)	Manual	Calibration	10	0.38	26.72	11.6	-1.06	24.16
		Evaluation	11	0.64	29.49	-13.9	-5.25	26.30
Chlorophyll-a (μ g L ⁻¹) (0–3 m)	Manual	Calibration	27	0.19	1.36	-54	-0.94	1.08
		Evaluation	14	0.12	1.11	-50	-2.46	0.95

Note. Each goodness of fit metric was calculated comparing model outputs and observational data for the time period (Calibration 2005–2009 or Evaluation 2010–2013) of interest.

3.2. Driver Data and Primary Response Variable Trends Over Simulation Period

Over the 31-year reference simulation, simulated June–September median and annual maximum chlorophyll-*a* concentration did not increase or decrease (Figure 4a), following historical records of Lake Sunapee (Steiner & Titus, 2017; NHDES, 2017): the temporal trend for simulated summer median chlorophyll-*a* had $R^2 = -0.02$, p = 0.18; annual maximum chlorophyll-*a* had $R^2 = -0.02$, p = 0.50). Simulated June–September median chlorophyll-*a* concentration ranged from 0.6 to 1.4 µg L⁻¹ (mean = 1.0 µg L⁻¹; SD = 0.23) over the 31 years. Simulated annual maximum chlorophyll-*a* concentration ranged from 1.2 to 2.2 µg L⁻¹ among years (mean = 1.7 µg L⁻¹; SD = 0.29).

There was an increasing trend in observed June–September mean air temperature and external phosphorus load from 1985 to 2015, but there was no significant change in annual maximum air temperature (Figures 4b and 4c). Observed June–September mean air temperature ranged from 15.9 to 18.9°C (mean = 17.5°C; SD = 0.74), and increased over the simulation period at a rate of 0.05° C yr⁻¹ ($R^2 = 0.42$, p < 0.001; Figure 4b). Observed annual maximum air temperature ranged from 26.4°C to 32.08°C (mean = 29.4°C; SD = 1.29; $R^2 = 0.10$, p = 0.50). Most external P entered the lake between March and September of each year. Total annual external P load ranged from 465 to 826 kg (mean = 674 kg, SD = 89.8), and increased over the simulation period at a rate of 4.2 kg yr⁻¹ ($R^2 = 0.18$, p = 0.02). March–September external phosphorus (P) load ranged from 252 to 499 kg (mean = 399 kg; SD = 55 kg), and increased at a rate of 2.2 kg yr⁻¹ ($R^2 = 0.14$, p = 0.04; Figure 4C).

3.3. Question 1: Reference Simulation Analysis—Relationships Between Annual Maximum Versus Median Chlorophyll-*a* Concentration and Climate and Catchment Load Drivers

Simulated annual maximum and summer median chlorophyll-*a* concentrations were sensitive to different drivers (Table 3). Simulated annual maximum chlorophyll-*a* concentration was the most strongly correlated with 3-year sum of external March–September TP load (Table 3; Spearman rho = 0.65, p < 0.001, n = 31). In contrast, simulated annual median chlorophyll-*a* concentration was most strongly correlated with June–September maximum air temperature (Table 3; Spearman rho = 0.67, p < 0.001, n = 31) (Table S2). A subset of the driver variables analyzed with the strongest response variable relationships and additional representative driver variables from each pathway are presented in Table 3; Spearman correlation results for all driver variables tested are listed in Table S2.





Figure 4. Time series of simulated summer maximum (circle, solid line) and median (triangle, dashed line) chlorophyll*a* concentrations aggregated on an annual scale in the surface (0–3 m) over the 31-year simulation (a); annual maximum (circle, solid line) and summer mean air (square, dashed line) temperature from North American Land Data Assimilation System (NLDAS-2) website (Xia et al., 2012) (b); and annual (circle, solid line) and March through September (square, dashed line) external phosphorus load (c).

3.4. Question 2: Scenario Analysis—Response of Maximum Versus Median Chlorophyll-*a* Concentrations to Projected Increases in Stream Inflow Nutrient Concentration and Air Temperature

Increases in combined nitrogen and phosphorus inflow concentrations had a stronger effect on both summer median and annual maximum chlorophyll-*a* concentrations than temperature or single nutrient additions alone (Figures 5 and 6). In each year of the 31-year simulation, summer median chlorophyll-*a* concentration was 0.9 to 2.1 times higher in Scenario 3: TNTP × 2 than in Scenario 4: $+2^{\circ}$ C (Figures 5 and 6). Similarly, annual maximum chlorophyll-*a* concentration was 1.1 to 2.5 times higher in Scenario 3: TNTP × 2 than in Scenario 4: $+2^{\circ}$ C (Figures 5 and 6). Further, summer median chlorophyll-*a* concentrations were 0.8 to 2.3 times higher in Scenario 3: TNTP × 2 than in Scenario 2: TP × 2 (Figures 5 and 6). Similarly, Scenario 3: TNTP × 2 than in Scenario 1: TN × 2 or Scenario 2: TP × 2 (Figures 5 and 6). Similarly, Scenario 3: TNTP × 2 exhibited annual maximum chlorophyll-*a* concentrations between 0.8 and 2.2 times higher than annual maximum chlorophyll-*a* concentrations in Scenario 1: TN × 2 or Scenario 1: TN × 2 or Scenario 5.



Table 3

Spearman Correlation Coefficients for Driver Variable Aggregations and Annual Maximum and Summer Median Chlorophyll-a

Category	Variable	Annual max Chl-a	Summer median Chl-a
Air Temperature (climate driver: Air Temperature	Mean of 3 years June–September maximum temperature	0.21	0.47***
Pathway in Figure 1)	Annual maximum temperature	0.29	0.67***
Total Nutrient Load (combined land-use +	Sum 3-yr March–September TP	0.62***	0.26
climate driver: Catchment Nutrient Pathway	Sum 3-yr March–May TP	0.58	0.22
& Precipitation Pathway in Figure 1)	Sum 3-water years TP	0.62***	0.27
	Sum 3-yr March–September TN	0.25	0.21
	Sum 3-water years TN	0.33	0.29
	March–May TP	0.48	0.10
	March–May TN	0.11	-0.06
Total Precipitation (climate driver:	Sum 3-yr March–May	0.09	0.03
Precipitation Pathway in Figure 1)	Sum 3-yr Jun–Sept	-0.35	0.21
	Sum 3-yr March–September	-0.25	0.19
	June–Sept	-0.35^{*}	0.22
	Water year	-0.25	0.17

Note. Three-year metrics include the year of response and previous 2 years. Bold text denotes significance of p < 0.001. Spearman correlation results with all driver variables are listed in Table S2. Water year is from October 1 through September 30.

Significance value of p < 0.05. Significance value of p < 0.01. Significance value of p < 0.001.



Figure 5. Time series of annual maximum and summer median chlorophyll-*a* for all scenarios (Reference = historical simulation; Scenario 1: $TN \times 2 = 2 \times total$ nitrogen external load; Scenario 2: $TP \times 2 = 2 \times total$ phosphorus external load; Scenario 3: $TPTN \times 2 = 2 \times total$ phosphorus and total nitrogen external load; Scenario 4 + 2°C = air temperature warming; Scenario 5: $TPTN \times 2 + 2°C = 2 \times total$ phosphorus and total nitrogen external load plus air temperature warming) for annual maximum chlorophyll-*a* concentrations (a) and summer median chlorophyll-*a* concentrations (b).





Figure 6. Distributions of 31-year annual maximum and median chlorophyll-*a* for five land use and climate scenarios (Reference = historical simulation; Scenario 1: $TN \times 2 = 2 \times total$ nitrogen external load; Scenario 2: $TP \times 2 = 2 \times total$ phosphorus external load; Scenario 3: $TPTN \times 2 = 2 \times total$ phosphorus and total nitrogen external load; Scenario 5: $TPTN \times 2 + 2^{\circ}C = 2 \times total$ phosphorus and total nitrogen external load; scenario 5: $TPTN \times 2 + 2^{\circ}C = 2 \times total$ phosphorus and total nitrogen external load; plus air temperature warming; a unique letter above a boxplot denotes a significantly different distribution from other scenarios (Bonferroni-corrected *p* < 0.002, Anderson-Darling test) for annual maximum chlorophyll-*a* concentrations (a) and summer median chlorophyll-*a* concentrations (b).

Warmer air temperature had a significant effect on summer median chlorophyll-*a* concentrations at both ambient and elevated nutrient inflow concentrations, but only increased annual maximum chlorophyll-*a* concentration at elevated nutrient inflow concentrations (Figure 6). Specifically, combined warmer air temperatures and increased nitrogen and phosphorus nutrient inflow concentrations (Scenario 5: TNTP × 2 + 2°C) increased both annual maximum and summer median chlorophyll-*a* concentrations in comparison to Scenario 3: TPTN × 2 (Bonferroni-corrected *p* < 0.002; Figure 6). However, warmer air temperatures (Scenario 4: +2°C) significantly affected summer median (Bonferroni-corrected *p* < 0.002; Figure 6b), but not annual maximum chlorophyll-*a* concentrations, in comparison to the reference





Figure 7. Spearman correlations between simulation output of driver variable (external TP load), mediating variables (surface water TP concentration, P uptake), and annual maximum chlorophyll-*a* concentration (a–c); between driver variable (air temperature), mediating variable (water temperature), and summer median chlorophyll-*a* concentration (d–e).

simulation (Bonferroni-corrected p = 0.01; Figure 6a). In the air temperature scenario (Scenario 4: +2°C), onset of thermal stratification was an average of 6 days earlier, thermocline depth was an average of 1.1 m lower, and summer surface water temperature was an average 1.3°C warmer than in the Reference simulation.

3.5. Confirming Driver Pathways

In the reference simulation analysis (section 3.3, Table 3), external P load was identified as the driver variable most associated with annual maximum chlorophyll-*a* concentration. We confirmed that the

Catchment Nutrient Pathway was the likely primary pathway by which external P load affected annual maximum chlorophyll-*a* concentration in the reference simulation, as indicated by the significant relationships between (1) total TP load and surface TP concentration (Figure 7a; Spearman rho = 0.95, p < 0.001), (2) surface TP concentration and phytoplankton P uptake (Figure 7b; Spearman rho = 0.93, p < 0.001), and (3) phytoplankton P uptake and annual maximum chlorophyll-*a* concentration (Figure 7c; Spearman rho = 0.80, p < 0.001).

In the reference simulation analysis (section 3.3), air temperature was identified as the driver most associated with summer median chlorophyll-*a* concentration (Table 3). We confirmed that the Air Temperature Pathway was the likely primary pathway by which air temperature affected summer median chlorophyll-*a* concentration response, as indicated by the significant relationship between (1) air temperature and water temperature (Figure 7d; Spearman rho = 0.72, p < 0.001), and (2) water temperature and summer median chlorophyll-*a* concentration (Figure 7e; Spearman rho = 0.49, p < 0.001).

4. Conclusions

By pairing long-term empirical data and a simulation model, our study revealed that climate and land-use drivers had different effects on median and maximum chlorophyll-*a* concentrations in Lake Sunapee, high-lighting the complexity of overall lake ecosystem responses to global change. Lake ecosystems have non-linear and interrelated biogeochemical processes, in which high variability in meteorological and nutrient load drivers can interact with phytoplankton dynamics to result in variable chlorophyll-*a* concentrations within and among years (Carey et al., 2016; Carpenter & Kitchell, 1987). Based on the results of our retrospective modeling and scenario analysis (section 3.3, Table 3), we focus on the Air Temperature Pathway and Catchment Nutrient Pathway in the discussion below (Figure 1), though we recognize the importance of the Precipitation Pathway (Collins et al., 2019), particularly as related to its interaction with the Catchment Nutrient Pathway through increased extreme precipitation events leading to sediment mobilization and increased nutrient loads to lakes (Carpenter et al., 2018; Sinha et al., 2017).

4.1. Time Scales of Influence

Our primary reference simulation result—that median chlorophyll-*a* concentration was significantly positively associated with *summer* air temperature while maximum chlorophyll-*a* concentration was significantly positively associated with *three-year* external P load (Table 3)—supports evidence that ecosystem drivers operate simultaneously on different time scales (*sensu*; Rinaldi & Scheffer, 2000). The Air Temperature Pathway (Figure 1) operates at daily and seasonal scales by altering the physical environment for phytoplankton through effects on water temperature (Liu et al., 2019), thermal stratification, mixing, and temperature-mediated biogeochemical reactions (Hamilton & Schladow, 1997). In contrast, given the 3-year residence time of Lake Sunapee, phytoplankton response to land-use change (Catchment Nutrient Pathway, Figure 1) operates over the course of years, resulting in time lags between land-use changes in the catchment, nutrient loads to the lake, and chlorophyll-*a* concentration. Consequently, it logically follows that chlorophyll-*a* concentration was differentially sensitive to these two pathways at different time scales.

In lake ecosystems and their catchments, as well as in GLM-AED model simulations, the pathways highlighted in Figure 1 are integrated, thereby resulting in ecological signals (e.g., chlorophyll-*a* concentration) with multiple time scales of variability. We found that annual maximum chlorophyll-*a* concentration in the reference simulation was most associated with 3-year external TP load (Table 3) and that increasing temperature alone (Scenario 4: $+2^{\circ}$ C) did not significantly change annual maximum chlorophyll-*a* concentration (Figure 6). The association of annual maximum chlorophyll-*a* concentration with the previous 3 years of external TP loading was in contrast to the expectation that peak phytoplankton biomass is most sensitive to within-year drivers (e.g., Batt et al., 2017). However, because annual maximum chlorophyll-*a* concentration did significantly increase in response to temperature at elevated TN and TP loads (compare Scenario 3 and Scenario 5 in Figure 6), it is possible that changes in multiyear drivers (e.g., optimal air temperatures) to promote high chlorophyll-*a* concentrations. This relationship could be an example of "accumulative carryover" (Ryo et al., 2019), in which longer-term increases in TN and TP loading create the conditions upon which optimal air temperature conditions are able to stimulate phytoplankton to a greater extent than if the 3-year TN and TP loads had been lower. While linking the year-to-year variability within a single simulation directly to the interaction of 3-year external TP load and daily air temperature is beyond the scope of our analysis, the divergence among Scenarios 3 and 5, but not between Scenario 4 and the reference simulation (Figure 6), supports our hypothesis of interactions between longer-term and shorter-term drivers (Rinaldi & Scheffer, 2000; Ryo et al., 2019). The relationship between drivers at different timescales could include lagged effects from extreme years (e.g., droughts and heat waves), though this is also beyond the scope of our analysis. However, taken together, our reference simulation analysis and scenario results suggest that the underlying causes of peak phytoplankton concentrations are likely a combination of daily and multiyear drivers (as outlined in Figure 1).

The ability to tease apart ecosystem drivers with different time scales of influence on chlorophyll-*a* concentration was only possible via a study that spanned many years. If we only analyzed 1 year of the 31-year reference simulation, our inferences about the drivers of chlorophyll-*a* concentrations in Lake Sunapee likely would not have been applicable to other years (Figure 4). Similarly, multiple years were needed to interpret the scenarios, as individual years exhibited different driver conditions and chlorophyll-*a* concentration responses (Figure 5). A sensitivity analysis (Figures S13 and S14) reveals that, on average, 15–20 simulation years were needed in the scenario distributions analysis to replicate the significant differences observed among all the scenarios for annual and summer chlorophyll-*a* concentration. The clarity of the long-term synoptic view apparent in the comparison of scenarios across 31 years (Figure 6) demonstrates the power of, and need for, long-term analysis for assessing complex ecosystem responses to drivers of change (e.g., Dodds et al., 2012; Hampton et al., 2019; Likens, 1988).

4.2. Underlying Mechanisms

The model simulation results provide insights into the potential mechanisms by which changes in air temperature and nutrient load drivers affected chlorophyll-*a* concentrations. Although GLM-AED is a deterministic model, it is not possible to trace an emergent lake ecosystem property, such as epilimnetic chlorophyll-*a* concentration, to its climate or land-use drivers through the model simulation equations because of the interdependency of model processes, high temporal and spatial (vertical) resolution of process interactions, and nonlinear dynamics. A common solution to this problem is to evaluate model output for broad relationships known to be ecologically relevant, such as nutrient availability and phytoplankton biomass, following Snortheim et al. (2017).

The strong associations between driver variables (e.g., external P load and air temperature), mediating variables that were simulated in the model (i.e., the variables that responded to changes in the driver variables and altered our focal response variables, such as epilimnetic P concentration and water temperature), and simulated focal response variables reveal the likely mechanistic driver pathways of chlorophyll-a concentration in Lake Sunapee (Figures 1 and 7; section 3.5). However, each driver pathway is not isolated from other potential pathways. For example, epilimnetic P concentration was modified in our scenarios via increased inflow stream P concentration (as a proxy for the Catchment Nutrient Pathway, e.g., chronic P additions from septic systems); however, increased erosion due to extreme precipitation could also increase inflow stream P concentration. Thus, our analysis is not intended to be exhaustive assessment of potential driver pathways of change in lake ecosystems, but rather to add to a growing understanding of the role of interacting driver pathways within the context of regional and local climate change (McCullough et al., 2019). While warming air temperatures will likely increase median chlorophyll-a concentrations in Lake Sunapee via warmer water temperatures (following Collins et al., 2019; Figure 7), warming air temperatures will also alter lake thermal stratification strength and timing (Kraemer et al., 2015; Woolway et al., 2019), and in turn, hypolimnetic dissolved oxygen concentrations and internal nutrient load dynamics (Marcé et al., 2010; North et al., 2014). Additionally, the interactive effect of multiple drivers of change (i.e., that temperature may have greater effect on annual maximum chlorophyll-a concentration at elevated TN and TP loads, Figure 6) highlights the need to understand how the pathways interact across a wide range of conditions.

The specific response of lake variables to pathways of change are mediated through lake-specific characteristics. For example, the Air Temperature Pathway can be mediated by DOC concentration, in which high DOC limits light penetration, thereby buffering the lake water temperature from rising air temperature (Read & Rose, 2013). The DOC-light penetration relationship also affects thermocline depth and is itself mediated by lake size, in which thermocline depth in larger lakes is less dependent on DOC-driven light limitation due to a stronger effect of wind-driven mixing of the epilimnion (Zwart et al., 2016). In sum, interacting pathways and lake-specific characteristics must be taken into account when interpreting the broad relationships observed in the Reference simulation and scenario testing model output.

Ecosystem models, in conjunction with observational data, reveal potential interactions that would be challenging to observe using empirical data alone. Emergent model simulation patterns inform our understanding of how lake ecosystems may function given our process-based knowledge. For example, in the scenario testing (section 3.4), annual maximum chlorophyll-*a* concentration increased more in the combined increased external TP and TN load scenario than in the increased external TP alone scenario (Figure 6), suggesting that co-limitation or serial limitation (Lewis et al., 2020) may play important roles governing chlorophyll-*a* concentrations in this ecosystem.

Dissolved inorganic nitrogen (DIN) to TP mass ratios in the scenarios demonstrate how Lake Sunapee's nutrient limitation may change in the future, following Bergström (2010)'s previous application of DIN: TP to identify limiting nutrients in oligotrophic lakes. In most years of the Reference simulation, DIN:TP is between 1.5 and 3.4 (Figure S15), a ratio at which co-limitation is expected (Bergström, 2010). In Scenario 1: TN × 2 and Scenario 2: TP × 2, the DIN:TP in most years shifts above 3.4 and below 1.5, indicating likely P-limitation and N-limitation, respectively, in response to nutrient addition. In the warming scenarios (Scenario 4: $+2^{\circ}$ C and Scenario 5: TPTN × 2 + 2° C), DIN:TP shifts toward N-limitation, in comparison to the Reference simulation and Scenario 3: TPTN × 2, respectively. Thus, it is not surprising that we observed co-limitation in the scenario results, which both supports previous studies conducted in other low-nutrient systems (Bergström, 2010; Lewis & Wurtsbaugh, 2008; Oleksy, 2019; Paerl et al., 2016; Saros et al., 2010) and informs future empirical work in Lake Sunapee.

Further, the significant difference in annual maximum chlorophyll-*a* concentration between Scenario 3: TPTN × 2 and Scenario 5: TPTN × 2 + 2°C and lack of significant difference between the reference simulation and Scenario 4: +2°C (Figure 6) indicate that temperature may have a larger effect on annual maximum chlorophyll-*a* concentration at higher TP concentrations. Separately, summer median chlorophyll-*a* concentration in the reference analysis was not associated with nutrient load, but significantly increased in Scenario 3: TPTN × 2, indicating that both the Air Temperature Pathway and Catchment Nutrient Pathway are important for summer median chlorophyll-*a* concentration response. These findings were made possible by our long-term ecosystem modeling approach with combined reference simulation analysis and scenario testing.

4.3. Modeling Oligotrophic Lake Dynamics

Our calibrated model tended to slightly overestimate total phosphorus concentrations and underestimate chlorophyll-*a* concentrations, potentially identifying limits to our ability to simulate phytoplankton dynamics in oligotrophic lakes. The phosphorus uptake and growth rate algorithms developed for GLM-AED may function better in mesotrophic or eutrophic lakes, where this model and similar other models have primarily been applied (e.g., Bruce et al., 2006; Gal et al., 2009; Kara et al., 2012; Snortheim et al., 2017). For example, phytoplankton may take up and store nutrients more rapidly than expected by current model algorithms due to their scavenging of organic N and P at lower inorganic nutrient concentrations (McKew et al., 2015). Further, warmer temperatures may facilitate increased nutrient uptake at low inorganic N to P ratios in oligotrophic lakes (Oleksy, 2019). Improving the representation of organic nutrient scavenging and other adaptations of phytoplankton taxa to oligotrophic conditions would potentially improve overall model fit of nutrient and chlorophyll-*a* concentrations in low-nutrient lakes.

Modeling low-nutrient systems such as Lake Sunapee is also challenging because lake nutrient samples are often near detection levels. Thus, water quality metrics often show few or no detectable patterns over time (Richardson et al., 2017). Although a water quality monitoring program has existed at Lake Sunapee for over 30 years (NHDES, 2017), most years have only three or four measurements of summer chlorophyll-*a* concentration, limiting our ability to validate hourly model output with observational data. Historical data sparsity emphasizes the importance of long-term modeling studies, where we can leverage limited observational data with our process understanding of ecosystem functioning to make inferences about ecosystem dynamics.

As oligotrophic freshwater systems become less common (Stoddard et al., 2016), it is important to understand a priori the role of climate and land use in driving lakes to a trophic state change. Thus, improving our ability to understand lake ecosystems when many metrics are at or below method detection limits is



critical. Future work in this area could include using mechanistic understanding from experiments to improve process-based models (e.g., Carpenter et al., 2011), paleoecological examination of lake dynamics (e.g., Jenny et al., 2015), and further long-term simulation modeling to tease apart the sensitivity of lakes to processes relevant in oligotrophic lake ecosystems.

4.4. Managing Oligotrophic Lakes

This study may inform lake monitoring programs. Our results indicate that median chlorophyll-*a* concentration may increase due to higher air temperature or nutrient load, but maximum chlorophyll-*a* will likely only increase with a higher nutrient load. Thus, the upper bound of chlorophyll-*a* concentrations is likely determined by the nutrient supply, which is generally expected in nutrient-limited oligotrophic lakes. Additionally, given that air temperatures are expected to increase in many regions due to climate change, maximum chlorophyll-*a* concentration is potentially a more sensitive metric in oligotrophic lakes for assessing nutrient load restrictions than median chlorophyll-*a* concentration.

Our work also highlights that oligotrophic lakes may be particularly sensitive to the combined threat of climate and land-use change. Similar to findings in other studies (e.g., Ho et al., 2019), there is mounting evidence that oligotrophic lakes are particularly sensitive to a changing climate separate from land-use-induced increases in nutrient load (Carey, Ibelings, et al., 2012; McCullough et al., 2019; Oleksy, 2019; Sadro et al., 2019). Our scenario analysis indicates that increased nutrient load in combination with higher air temperature had the greatest effect on chlorophyll-*a* concentration (Figure 6), furthering the evidence of the dual threat of climate and land-use change. This is similar to other studies finding interactive effects of climate and other eutrophication drivers (Jeppesen et al., 2010; Moss et al., 2011; Oleksy, 2019; Trolle et al., 2011). Therefore, it is likely that nutrient load reduction targets to achieve water quality goals should be even lower in a changing climate, following earlier work by Jeppesen et al. (2009), Brookes and Carey (2011), and Kosten et al. (2012).

Oligotrophic lake ecosystems provide essential ecosystem services (Batabyal et al., 2003), which makes it imperative to figure out how to maintain oligotrophic lake systems into the future in the face of their decline (Stoddard et al., 2016). Two main findings from this study provide context for managing oligotrophic lakes: (1) the differential response of maximum versus median chlorophyll-*a* concentration to nutrient loading and air temperature highlights the need to consider different aggregation metrics when assessing oligotrophic lake ecosystem response to changing drivers, and (2) the interaction between air temperature and nutrient load imply that even more ambitious nutrient load reductions may be required in a warming climate to achieve water quality goals. Given the rapid loss of oligotrophic lakes and the potential for interactive threats associated with climate and land-use change, this study sets the stage for future research and provides useful perspective on how to manage oligotrophic lakes in the face of climate and land-use change.

Data Availability Statement

Data sets for this research are available in these in-text data citation references: Ward et al. (2020), (CC BY license, public access available at environmentaldatainitiative.org), Richardson et al. (2020), (CC BY license, public access available at environmentaldatainitiative.org), Ewing et al. (2020), (CC BY license, public access available at environmentaldatainitiative.org), Cottingham (2020) (CC BY license, public access available at environmentaldatainitiative.org), Cottingham (2020) (CC BY license, public access available at environmentaldatainitiative.org), Cottingham (2020) (CC BY license, public access available at environmentaldatainitiative.org), Cottingham (2020) (CC BY license, public access available at environmentaldatainitiative.org), Xia et al. (2012), (U.S. Public Domain, public access available at disc.gsfc. nasa.gov/), MRLC (2018), (U.S. Public Domain, public access available at mrlc.gov/), CSRC (2002) (no license information provided, public access available at granit.unh.edu/). Models used in this research are available online (http://aed.see.uwa.edu.au/research/models/GLM/ and https://github.com/AquaticEco Dynamics/GLM).

Conflict of Interest

The authors declare no real or perceived conflicts of interest.

References

Abatzoglou, J. T. (2013). Development of gridded surface meteorological data for ecological applications and modelling. *International Journal of Climatology*, *33*, 121–131. https://doi.org/10.1002/joc.3413

Acknowledgments

We give a special thanks to Robert Wood and June Fichter from the Lake Sunapee Protective Association for data collection, curation, and extensive research support over the course of the study. We thank Isabella Oleksy, GLEON colleagues, the Carey Lab at Virginia Tech, and two anonymous reviewers for providing constructive feedback. We thank the many field and laboratory assistants who contributed to the Lake Sunapee Protective Association (LSPA) Long Term Monitoring (LMP) database and collection of stream data. This work was financially supported by National Science Foundation grants EF-0842267, EF-0842112, EF-0842125, ICER-1517823, DEB-1753639, and CNS-1737424; NIH P20GM103506, the William H. Neukom 1964 Institute for Computational Science at Dartmouth, the Lake Sunapee Protective Association, the Frey Foundation, and internal funding from Dartmouth College, Bates College, and Virginia Tech.

Arvola, L., Einola, E., & Järvinen, M. (2015). Landscape properties and precipitation as determinants for high summer nitrogen load from boreal catchments. *Landscape Ecology*, 30, 429–442. https://doi.org/10.1007/s10980-015-0166-8

Batabyal, A. A., Kahn, J. R., & O'Neil, R. V. (2003). On the scarcity value of ecosystem services. Environmental Economics and Management, 46(2003), 334–352. https://doi.org/10.1016/S0095-0696(02)00040-2

Batt, R. D., Carpenter, S. R., & Ives, A. R. (2017). Extreme events in lake ecosystem time series. *Limnology and Oceanography Letters*, *2*, 63–69. https://doi.org/10.1002/lol2.10037

Bennett, N. D., Croke, B. F. W., Gauriso, G., Guillaume, J. H. A., Hamilton, S. H., Jakeman, A. J., et al. (2013). Characterising performance of environmental models. *Environmental Modelling & Software*, 40, 1–20. https://doi.org/10.1016/j.envsoft.2012.09.011

Bergström, A. (2010). The use of TN:TP and DIN:TP ratios as indicators for phytoplankton nutrient limitation in oligotrophic lakes affected by N deposition. *Aquatic Sciences*, 72, 277–281. https://doi.org/10.1007/s00027-010-0132-0

Bormann, F. H., Likens, G. E., Siccama, T. G., Pierce, R. S., & Eaton, J. S. (1974). The export of nutrients and recovery of stable conditions following deforestation at Hubbard Brook. *Ecological Monographs*, 44(3), 255–277. https://doi.org/10.2307/2937031

Brookes, J. D., & Carey, C. C. (2011). Resilience to blooms. Science, 334, 46-47. https://doi.org/10.1126/science.1207349

Bruce, L. C., Hamilton, D., Imberger, J., Gal, G., Gophen, M., Zohary, T., & Hambright, K. D. (2006). Numerical simulation of the role of zooplankton in C, N, and P cycling in Lake Kinneret, Israel. *Ecological Modelling*, 193, 412–436. https://doi.org/10.1016/j. ecolmodel.2005.09.008

Bruesewitz, D. A., Carey, C. C., Richardson, D. C., & Weathers, K. C. (2015). Under-ice thermal stratification dynamics of a large, deep lake revealed by high-frequency data. *Limnology and Oceanography*, 60, 347–359. https://doi.org/10.1002/lno.10014

Bucak, T., Trolle, D., Tavşanoğlu, Ü. N., Çakıroğlu, A. İ., Özen, A., Jeppesen, E., & Beklioğlu, M. (2018). Modeling the effects of climatic and land use changes on phytoplankton and water quality of the largest Turkish freshwater lake: Lake Beyşehir. Science of the Total Environment, 621, 802–816. https://doi.org/10.1016/j.scitotenv.2017.11.258

Carey, C. C., Cottingham, K. L., Weathers, K. C., Brentrup, J. A., Ruppertsberger, N. M., Ewing, H. A., & Hairston, N. G. Jr. (2014). Experimental blooms of the cyanobacterium *Gloeotrichia echinulata* increase phytoplankton biomass, richness and diversity in an oligotrophic lake. *Journal of Plankton Research*, 36, 364–377. https://doi.org/10.1093/plankt/fbt105

Carey, C. C., Ewing, H. A., Cottingham, K. L., Weathers, K. C., Thomas, R. Q., & Haney, J. F. (2012). Occurrence and toxicity of the cyanobacterium *Gleotrichia echinulata* in low-nutrient lakes in the northeastern United States. *Aquatic Ecology*, 46, 395–409. https://doi. org/10.1007/s10452-012-9409-9

Carey, C. C., Hanson, P. C., Lathrop, R. C., & St. Amand, A. L. (2016). Using wavelet analyses to examine variability in phytoplankton seasonal succession and annual periodicity. *Journal of Plankton Research*, 38, 27–40. https://doi.org/10.1093/plankt/fbv116

Carey, C. C., Ibelings, B. W., Hoffmann, E. P., Hamilton, D. P., & Brookes, J. D. (2012). Eco-physiological adaptations that favour freshwater cyanobacteria in a changing climate. Water Research, 46, 1394–1407. https://doi.org/10.1016/j.watres.2011.12.016

Carey, C. C., Weathers, K. C., Ewing, H. A., Greer, M. L., & Cottingham, K. L. (2014). Spatial and temporal variability in recruitment of the cyanobacterium *Gloeotrichia echinulata* in an oligotrophic lake. *Freshwater Science*, 33, 577–592. https://doi.org/10.1086/ 675734

Carlson, R. E. (1977). A trophic state index for lakes. Limnology and Oceanography, 22(2), 361–369. https://doi.org/10.4319/ lo.1977.22.2.0361

Carpenter, S. R., Booth, E. G., & Kucharik, C. J. (2018). Extreme precipitation and phosphorus loads from two agricultural watersheds. Limnology and Oceanography, 63, 1221–1233. https://doi.org/10.1002/lno.10767

Carpenter, S. R., & Kitchell, J. F. (1987). The temporal scale of variance in limnetic primary production. *The American Naturalist*, 129(3), 417–433. https://doi.org/10.1086/284646

Cobourn, K. M., Carey, C. C., Boyle, K. J., Duffy, C., Dugan, H. A., Farrell, K. J., et al. (2018). From concept to practice to policy: Modeling coupled natural and human systems in lake catchments. *Ecosphere*, 9, e02209. https://doi.org/10.1002/ecs2.2209

Collins, S. M., Yuan, S., Tan, P. N., Oliver, S. K., Lapierre, J. F., Cheruvelil, K. S., et al. (2019). Winter precipitation and summer temperature predict lake water quality at macroscales. *Water Resources Research*, 55, 2708–2721. https://doi.org/10.1029/ 2018WR023088

Complex Systems Research Center. (2002). New Hampshire land cover assessment, accessed at <URL:http://www.granit.sr.unh.edu>.

Cottingham, K. L. (2020). Surface pelagic total nitrogen concentrations in Lake Sunapee, USA 2009 to 2012. Environmental data initiative. https://doi.org/10.6073/pasta/57282aff9a884449864399012e36abf2. Dataset accessed 5/15/2020.

Couture, R. M., Moe, S. J., Lin, Y., Kaste, O., Haande, S., & Solheim, A. L. (2018). Simulating water quality and ecological status of Lake Vansjø, Norway, under land-use and climate change by linking process-oriented models with a Bayesian network. *Science of the Total Environment*, 621, 713–724. https://doi.org/10.1016/j.scitotenv.2017.11.303

Couture, R. M., Tominaga, K., Starrfelt, J., Moe, S. J., Kaste, O., & Wright, R. F. (2014). Modelling phosphorus loading and algal blooms in a Nordic agricultural catchment-lake system under changing land-use and climate. *Environmental Science: Processes & Impacts*, 16, 1588–1599. https://doi.org/10.1039/c3em00630a

De Senerpont Domis, L. N., Elser, J. J., Gsell, A. S., Huszar, V. L. M., Ibelings, B. W., Jeppesen, E., et al. (2013). Plankton dynamics under different climatic conditions in space and time. *Freshwater Biology*, 58, 463–482. https://doi.org/10.1111/fwb.12053

Dodds, W. K., Robinson, C. T., Gaiser, E. E., Hansen, G. J. A., Powell, H., Smith, J. M., et al. (2012). Surprises and insights from long-term aquatic data sets and experiments. *BioScience*, 62, 709–721. https://doi.org/10.1525/bio.2012.62.8.4

Ewing H. A., Weathers K. C., & Steele, B. G. (2020). High resolution stream temperature, pressure, and estimated depth from transducers in six streams in the Lake Sunapee watershed, New Hampshire, USA 2010–2014. Environmental Data Initiative. https://doi.org/10.6073/ pasta/12a18d9d178b17d2ac333930f3efca9f. Dataset accessed 2/07/2020.

Gal, G., Hipsey, M. R., Parparov, A., Wagner, U., Makler, V., & Zohary, T. (2009). Implementation of ecological modeling as an effective management and investigation tool: Lake Kinneret as a case study. *Ecological Modelling*, 220, 1697–1718. https://doi.org/10.1016/j. ecolmodel.2009.04.010

Hamilton, D. P., & Schladow, S. G. (1997). Prediction of water quality in lakes and reservoirs. Part I—Model description. Ecological Modelling, 96(1997), 91–110. https://doi.org/10.1016/S0304-3800(96)00062-2

Hampton, S., Scheuerell, M. D., Church, M. J., & Melack, J. M. (2019). Long-term perspectives in aquatic research. *Limnology and Oceanography*, 64, S2–S10. https://doi.org/10.1002/lno.11092

Hayes, N. M., Vanni, M. J., Horgan, M. J., & Renwick, W. H. (2015). Climate and land use interactively affect lake phytoplankton nutrient limitation status. *Ecology*, 96, 392–402. https://doi.org/10.1890/13-1840.1

Carpenter, S. R., Cole, J. J., Pace, M. L., Batt, R., Brock, W. A., Cline, T., et al. (2011). Early warnings of regime shifts: A whole-ecosystem experiment. *Science*, 332, 1079–1082. https://doi.org/10.1126/science.1203672

Hipsey, M. R., Bruce, L. C., Boon, C., Busch, B., Carey, C. C., Hamilton, D. P., et al. (2019). A general lake model (GLM 3.0) for linking with high-frequency sensor data from the Global Lake Ecological Observatory Network (GLEON). *Geoscientific Model Development*, 12, 473–523. https://doi.org/10.5194/gmd-12-473-2019

Ho, J. C., Michalak, A. M., & Pahlevan, N. (2019). Widespread global increase in intense lake phytoplankton blooms since the 1980s. *Nature*, 574, 667–670. https://doi.org/10.1038/s41586-019-1648-7

- Jenny, J. P., Francus, P., Normandeau, A., LaPointe, F., Perga, M. E., Ojala, A., et al. (2015). Global spread of hypoxia in freshwater ecosystems during the last three centuries is caused by rising local human pressure. *Global Change Biology*, 22, 1481–1489. https://doi.org/ 10.1111/gcb.13193
- Jeppesen, E., Kronvang, B., Meerhoff, M., Sondergaard, M., Hansen, K. M., Andersen, H. E., et al. (2009). Climate change effects on runoff, catchment phosphorus loading and lake ecological state, and potential adaptations. *Journal of Environmental Quality*, *38*, 1930–1941. https://doi.org/10.2134/jeq2008.0113
- Jeppesen, E., Moss, B., Bennion, H., Carvalho, L., De Meester, L., Feuchtmayr, H., Friberg, N., Gessner, M. O., Hefting, M., Lauridsen, T. L., Liboriussen, L., Malmquist, H. J., May, L., Meerhoff, M., Olafsson, J. S., Soons, M. B., & Verhoeven, J. T. A., (2010). Interaction of climate change and eutrophication. In M. Kernan, R. W. Battarbee, & B. Moss (Eds.), *Climate change impacts on freshwater ecosystems* (pp. 119–151). Chinchester (UK): Wiley Blackwell.
- Jones, R. A., Rast, W., & Lee, G. F. (1979). Relationship between summer mean and maximum chlorophyll-*a* concentrations in lakes. *Environmental Science and Technology*, 13(7), 869–870. https://doi.org/10.1021/es60155a018
- Kara, E. K., Hanson, P., Hamilton, D., Hipsey, M. R., McMahon, K. D., Read, J. S., et al. (2012). Time-scale dependence in numerical simulations: Assessment of physical, chemical, and biological predictions in a stratified lake at temporal scales of hours to months. *Environmental Modelling & Software*, 35, 104–121. https://doi.org/10.1016/j.envsoft.2012.02.014

King, K. W., Balogh, J. C., & Harmel, R. D. (2007). Nutrient flux in storm water runoff and baseflow from managed turf. Environmental Pollution, 150, 321–328. https://doi.org/10.1016/j.envpol.2007.01.038

- Kosten, S., Huszar, V. L. M., Becares, E., Costa, L. S., van Donk, E., Hansson, L. A., et al. (2012). Warmer climates boost cyanobacterial dominance in shallow lakes. *Global Change Biology*, 18, 118–126. https://doi.org/10.1111/j.1365-2486.2011.02488.x
- Kraemer, B. M., Anneville, O., Chandra, S., Dix, M., Kuusisto, E., Livingstone, D. M., et al. (2015). Morphometry and average temperature affect lake stratification responses to climate change. *Geophysical Research Letters*, 42, 4981–4988. https://doi.org/10.1002/ 2015GL064097
- Lewis, A. S. L., Kim, B. S., Edwards, H. L., Wander, H. L., Garfield, C. M., Murphy, H. E., et al. (2020). Prevalence of phytoplankton limitation by both nitrogen and phosphorus related to nutrient stoichiometry, land use, and primary producer biomass across the northeastern United States. *Inland Waters*, 10, 42–50. https://doi.org/10.1080/20442041.2019.1664233

Lewis, W. M., & Wurtsbaugh, W. A. (2008). Control of lacustrine phytoplankton by nutrients: Erosion of the phosphorus paradigm. *International Review of Hydrobiology*, 93, 446–465. https://doi.org/10.1002/iroh.200811065

- Likens, G. E. (1988). Long-term studies in ecology, approaches and alternatives. New York: Springer-Verlag.
- Liu, X., Feng, J., & Wang, Y. (2019). Chlorophyll a predictability and relative importance of factors governing lake phytoplankton at different timescales. Science of the Total Environment, 648, 472–480. https://doi.org/10.1016/j.scitotenv.2018.08.146
- LSPA, Steele, B., & Weathers, K. C. (2018). Lake Sunapee high frequency weather data measured on buoy—2007–2017. Environmental Data Initiative. https://doi.org/10.6073/pasta/175cfde22aaeee7349285d2c9fd298d9. Dataset accessed 12/16/2019.
- Magee, M. R., McIntyre, P. B., Hanson, P. C., & Wu, C. H. (2019). Drivers and management implications of long-term cisco oxythermal habitat decline in Lake Mendota, WI. Environmental Management, 63, 396–407. https://doi.org/10.1007/s00267-018-01134-7
- Marcé, R., Rodríguez-Arias, M. À., Garcías, J. C., & Armengol, J. (2010). El Niño Southern Oscillation and climate trends impact reservoir water quality. *Global Change Biology*, *16*, 2857–2865. https://doi.org/10.1111/j.1365-2486.2010.02163.x
- McCarthy, J. (2008). New Hampshire Stormwater Manual: Volume 1 Stormwater and Antidegradation. Comprehensive Environmental Inc, Ed. Accessed online 23 Jan 2020 at https://www.des.nh.gov/organization/commissioner/pip/publications/wd/documents/wd-08-20a.pdf
- McCullough, I. M., Cheruvelil, K. S., Collins, S. M., & Soranno, P. A. (2019). Geographic patterns of the climate sensitivity of lakes. *Ecological Applications*, 29, e01836. https://doi.org/10.1002/eap.1836
- McKew, B. A., Metodieva, G., Raines, C. A., Metodiev, M. V., & Geider, R. J. (2015). Acclimation of *Emiliania hyxleyi* (1516) to nutrient limitation involves precise modification of the proteome to scavenge alternative sources of N and P. *Environmental Microbiology*, 17, 4050–4062. https://doi.org/10.1111/1462-2920.12957
- Meyer, J. L., Sale, M. J., Mulholland, P. J., & Poff, N. L. (1999). Impacts of climate change on aquatic ecosystem functioning and health. Journal of the American Water Resources Association, 35(6), 1373–1386. https://doi.org/10.1111/j.1752-1688.1999.tb04222.x
- Moss, B., Kosten, S., Meerhoff, M., Battarbee, R. W., Jeppesen, E., Mazzeo, N., et al. (2011). Allied attack: Climate change and eutrophication. *Inland Waters*, 1, 101–105. https://doi.org/10.5268/IW-1.2.359
- NHDES: New Hampshire Department of Environmental Services. (2017). Volunteer Lake Assessment Program Individual Lake Reports: Sunapee Lake, Sunapee, NH. Accessed at https://www.des.nh.gov/organization/divisions/water/wmb/vlap/annual_reports/2017/lake-reports.htm
- North, R. P., North, R. L., Livingstone, D. M., Koster, Ö., & Kipfer, R. (2014). Long-term changes in hypoxia and soluble reactive phosphorus in the hypolimnion of a large temperate lake: consequences of a climate regime shift. *Global Change Biology*, 20, 811–823. https:// doi.org/10.1111/gcb.12371
- O'Reilly, C. M., Sharma, S., Gray, D. K., Hampton, S. E., Read, J. S., Rowley, R. J., et al. (2015). Rapid and highly variable warming of lake surface waters around the globe. *Geophysical Research Letters*, 42, 10,773–10,781. https://doi.org/10.1002/2015GL066235
- Oleksy, I. A. (2019). Algal blooms in the alpine: Investigating the coupled effects of warming and nutrient deposition on mountain lakes, (doctoral dissertation). Retrieved from Mountain Scholar. (https://mountainscholar.org/bitstream/handle/10217/197430/Oleksy_ colostate_0053A_15661.pdf?sequence=1&isAllowed=y). Fort Collins, CO: Colorado State University.
- Paerl, H. W., & Huisman, J. (2008). Blooms like it hot. *Science*, 320, 57–58. https://doi.org/10.1126/science.1155398
- Paerl, H. W., Scott, J. T., McCarthy, M. J., Newell, S. E., Gardner, W. S., Havens, K. E., et al. (2016). It takes two to tango: When and where dual nutrient (N & P) reductions are needed to protect lakes and downstream ecosystems. *Environmental Science & Technology*, 50(20), 10,805–10,813. https://doi.org/10.1021/acs.est.6b02575
- Piccolroaz, S., Toffolon, M., & Majone, B. (2015). The role of stratification on lakes' thermal response: The case of Lake Superior. Water Resources Research, 51, 7878–7894. https://doi.org/10.1002/2014WR016555
- Pick, F. R. (2015). Blooming algae: The Canadian perspective on the rise of toxic cyanobacteria. Canadian Journal of Fisheries and Aquatic Sciences, 73, 1149–1158.

- R Core Team (2017). An introduction to R. In R: A language and environment for statistical computing, (2–5). Vienna, Austria: R Core Team. https://www.r-project.org/
- Razali, N. M., & Wah, Y. B. (2011). Power comparisons of Shapiro-Wilk, Kolmogorov-Smirnov, Lilliefors and Anderson-Darling tests. Journal of Statistical Modeling and Analytics, 2, 21–33.
- Read, J. S., & Rose, K. C. (2013). Physical responses of small temperate lakes to variation in dissolved organic concentrations. *Limnology and Oceanography*, 58, 921–931. https://doi.org/10.4319/lo.2013.58.3.0921
- Richardson, D. C., Carey, C. C., Bruesewitz, D. A., & Weathers, K. C. (2017). Intra- and inter-annual variability in metabolism in an oligotrophic lake. Aquatic Sciences, 79, 319–333. https://doi.org/10.1007/s00027-016-0499-7
- Richardson, D. C., Carey, C. C., Weathers, K. C., Bruesewitz, D. A., LSPA, & Steele, B. (2020). High frequency meteorological, drift-corrected dissolved oxygen, and thermistor temperature data—Lake Sunapee Buoy, NH, USA, 2007–2013. Environmental Data Initiative. https://doi.org/10.6073/pasta/846a36bf6fd704e508511e5f8a2ab3b5. Dataset accessed 2/05/2020.
- Rigosi, A., Carey, C. C., Ibelings, B. W., & Brookes, J. D. (2014). The interaction between climate warming and eutrophication to promote cyanobacteria is dependent on trophic state and varies among taxa. *Limnology and Oceanography*, 59, 99–114. https://doi.org/10.4319/ lo.2014.59.1.0099
- Rinaldi, S., & Scheffer, M. (2000). Geometric analysis of ecological models with slow and fast processes. *Ecosystems*, 3(6), 507–521. https://doi.org/10.1007/s100210000045
- Ryo, M., Aguilar-Trigueros, C. A., Pinek, L., Muller, L. A. H., & Rilling, M. C. (2019). Basic principles of temporal dynamics. Trends in Ecology & Evolution, 34, 723–733. https://doi.org/10.1016/j.tree.2019.03.007
- Sadro, S., Melack, J. M., Sickman, J. O., & Skeen, K. (2019). Climate warming response of mountain lakes affected by variations in snow. Limnology and Oceanography Letters, 4, 9–17. https://doi.org/10.1002/lol2.10099
- Saros, J. E., Rose, K. C., Clow, D. W., Stephens, V. C., Nurse, A. B., Arnett, H. A., et al. (2010). Melting alpine glaciers enrich high-elevation lakes with reactive nitrogen. *Environmental Science and Technology*, 44, 4891–4896. https://doi.org/10.1021/es100147j
- Schloss, J. (1990). Lake Sunapee nutrient budget study. Durham, New Hampshire, USA: University of New Hampshire. https://extension.unh.edu/resources/files/Resource005326_Rep7480.pdf
- Sinha, E., Michalak, A. M., & Balaji, V. (2017). Eutrophication will increase during the 21st century as a result of precipitation changes. *Science*, 357, 405–408. https://doi.org/10.1126/science.aan2409
- Snortheim, C. A., Hanson, P. C., McMahon, K. D., Read, J. S., Carey, C. C., & Dugan, H. A. (2017). Meteorological drivers of hypolimnetic anoxia in a eutrophic, north temperate lake. *Ecological Modelling*, 343, 39–53. https://doi.org/10.1016/j.ecolmodel.2016.10.014
- Solomon, C. T., Bruesewitz, D. A., Richardson, D. C., Rose, K. C., van de Bogert, M. C., Hanson, P. C., et al. (2013). Ecosystem respiration: Drivers of daily variability and background respiration in lakes around the globe. *Limnology and Oceanography*, 58, 849–866. https://doi. org/10.4319/lo.2013.58.3.0849
- Steiner S., & Titus, M. (2017). NH Department of Environmental Services (NHDES) Watershed Management Bureau Biology Section: Volunteer Lake Assessment Program (VLAP) Environmental Monitoring data 1995–2014. Environmental data initiative. https://doi.org/ 10.6073/pasta/1df3238af71108db248f667b9561023d. Dataset accessed 1/23/2020.
- Stoddard, J. L., Van Sickle, J., Herlihy, A. T., Brahney, J., Paulsen, S., Peck, D. V., et al. (2016). Continental-scale increase in lake and stream phosphorus: Are oligotrophic systems disappearing in the United States? *Environmental Science & Technology*, 50, 3409–3415. https:// doi.org/10.1021/acs.est.5b05950
- Sunapee Area Watershed Coalition, (2008). Management Plan for the Lake Sunapee Watershed. Retrieved from https://www.des.nh.gov/ organization/divisions/water/wmb/was/documents/lake_sunapee_wbp.pdf
- Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. *American Meteorological Society*, 93, 485–498. https://doi.org/10.1175/BAMS-D-11-00094.1
- Trolle, D., Elliot, J. A., Mooij, W. M., Janse, J. H., Bolding, K., Hamilton, D. P., & Jeppesen, E. (2014). Advancing projections of phytoplankton responses to climate change through ensemble modelling. *Environmental Modelling & Software*, 61, 371–379. https://doi.org/ 10.1016/j.envsoft.2014.01.032
- Trolle, D., Hamilton, D. J., Pilditch, C. A., Duggan, I. C., & Jeppesen, E. (2011). Predicting the effects of climate change on trophic status of three morphologically varying lakes: Implications for lake restoration and management. *Environmental Modeling Software*, 26, 354–370. https://doi.org/10.1016/j.envsoft.2010.08.009
- Turner, M. G., & Carpenter, S. R. (2017). Ecosystem modeling for the 21st Century. *Ecosystems*, 20, 211–214. https://doi.org/10.1007/s10021-016-0076-x
- USCOE (United States Corps of Engineers) (1998). *Runoff from snowmelt, Engineer Manual 1110-2-1406*. Washington DC. Retrieved from: US Army Corps of Engineers. https://www.publications.usace.army.mil/Portals/76/Publications/EngineerManuals/EM_1110-2-1406. pdf
- Van Mullem, J. A., Garen, D., Woodward, D., & Mockus, V. (2004). Chapter 11: Snowmelt, Part 630 Hydrology National Engineering Handbook. Washington, DC. Retrieved from: United States Department of Agriculture. https://www.wcc.nrcs.usda.gov/ftpref/wntsc/ H&H/NEHhydrology/ch11.pdf
- Ward, N. K. Steele, B., Carey, C. C., & Hanson, P. C. (2020). Lake Sunapee parameter set and driver files for the General Lake Model Aquatic EcoDynamics (GLM AED). *Environmental Data Initiative*. https://doi.org/10.6073/pasta/681f48dcc63bb43655de685891f414e6. Dataset accessed 2/05/2020.
- Weathers, K. C., Strayer, D. L., & Likens, G. E. (2013). Introduction to ecosystem science. In Fundamentals of ecosystem science, (3–23). San Diego, CA: Academic Press. https://doi.org/10.1016/B978-0-08-091680-4.00001-9
- Wetzel, R. G. (2001). The phosphorus cycle & the nitrogen cycle. In *Limnology, lake and river ecosystems* (pp. 239–288). San Diego, CA: Academic Press.
- Williamson, C. E., Dodds, W., Kratz, T. K., & Palmer, M. A. (2008). Lakes and streams as sentinels of environmental change in terrestrial and atmospheric processes. Frontiers in Ecology and the Environment, 6, 247–254. https://doi.org/10.1890/070140
- Winter, J. G., DeSellas, A. M., Fletcher, R., Heintsch, L., Morley, A., Nakamoto, L., & Utsumi, K. (2011). Algal blooms in Ontario, Canada: Increases in reports since 1994. *Lake and Reservoir Management*, 27, 107–114. https://doi.org/10.1080/07438141.2011.557765
- Woolway, R. I., Weyhenmeyer, G. A., Schmid, M., Dokulil, M., de Eyto, E., Maberly, S. C., et al. (2019). Substantial increase in minimum lake surface temperatures under climate change. *Climatic Change*, 155, 81–94. https://doi.org/10.1007/s10584-019-02465-y
- Xia, Y., Mitchell, K., Ek, M., Sheffield, J., Cosgrove, B., Wood, E., Luo, L, Alonge, C., Wei, H., Meng, J., Livneh, B., Lettenmaier, D., Koren, V., Duan, Q., Mo, K., Fan, Y., & Mocko, D. (2012). Continental-scale water and energy flux analysis and validation for the North American Land Data Assimilation System project phase 2 (NLDAS-2): 1. Intercomparison and application of model products. *Journal of Geophysical Research: Atmospheres*, 117(D3), https://doi.org/10.1029/2011jd016048



Zambrana-Bigiarini, M. (2017). hydroGOF: Goodness-of-fit functions for comparison of simulated and observed hydrological time series. R Package Version 0.3 – 10. Zwart, J. A., Craig, N., Kelly, P. T., Sebestyen, S. D., Solomon, C. T., Weidel, B. C., & Jones, S. E. (2016). Metabolic and physiochemical

Zwart, J. A., Craig, N., Kelly, P. T., Sebestyen, S. D., Solomon, C. T., Weidel, B. C., & Jones, S. E. (2016). Metabolic and physiochemical responses to a whole-lake experimental increase in dissolved organic carbon in a north-temperate lake. *Limnology and Oceanography*, 61, 723–734. https://doi.org/10.1002/lno.10248